Author's response to reviews

Title: Efficacy of a new absorbent pad on incontinence-associated dermatitis in older women: cluster randomized controlled trial

Authors:

Junko Sugama (junkosgm@mhs.mp.kanazawa-u.ac.jp)
Yoshie Shigeta (yshigeta@mhs.mp.kanazawa-u.ac.jp)
Chizuko Konya (konya@kanazawa-med.ac.jp)

Version: 2 Date: 29 August 2011

Author's response to reviews:

30-August-2011

Dear editor;

According to your suggestion, we got the trial registration ID and described it in the abstract. And this manuscript was given the professional editing service twice by Edanz which recommended from the journal editorial office, BioMed Central.

On behalf of all the authors, I would like to ask you to consider our manuscript entitled "Effectiveness of a new absorbent pad on incontinence-associated dermatitis in older adult women: A cluster randomized controlled trial" for publication in BioMed Geriatrics as an original research article.

with the drafting of the article.

Junko Sugama, PhD, RN

Department of Clinical Nursing, Institute of Medical, Pharmaceutical and Health Sciences, Kanazawa University, 5-11-80 Kodatsuno, Kanazawa, 920-0942 Ishikawa, Japan,
Tel +81-76-265-2555 Fax +81-76-265-2555
E-mail: junkosgm@mhs.mp.kanazawa-u.ac.jp